Author’s response to reviews

Title: Gamma-tocotrienol treatment increased peroxiredoxin-4 expression in HepG2 liver cancer cell line

Authors:

Farahani Abdul Rahman Sazli (farahsazli@gmail.com)
Zakiah Jubri (zakiah@medic.ukm.my)
Mariati Abdul Rahman (mariati_ar@dental.ukm.my)
Saiful Anuar Karsani (saiful72@um.edu.my)
Abdul Gapor Md Top (gapor@mpob.gov.my)
Wan Zurinah Wan Ngah (zurina@medic.ukm.my)

Version: 5  Date: 10 February 2015

Author’s response to reviews:

Dear Editor,

RESPONSE TO THE REVIEWER CONCERNS FOR THE REVISED MANUSCRIPT – Gamma-tocotrienol treatment increased peroxiredoxin-4 expression in HepG2 liver cancer cell line

Herein, I would like to response on reviewers concern as stated below:

Reviewer: Hiromi Sato
Reviewer’s report:
Major Compulsory Revisions

1. Authors stated regarding Table 2 (in his first review query No. 2) he totally understand what was indicated by the authors. To present the fact objectively, he suggested to put “+” expression seems better than arrows if it could not be put a numerical value and we might want to show all or nothing detected in only one of the samples. That might be more natural and acceptable to readers.

Response:

We agree with the suggestion and have changed accordingly.

2. In Figure 4 legend, the reviewer suggested to put described what statistical method was used. It is better to add the description in Materials and Methods section.

Response:

We agree and have added in Materials and Methods.

Thank you
Best regards

DR ZAKIAH JUBRI

Corresponding author